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07 September 2010
MEMORANDUM FOR CENWP-OP, ATTN: Operations Biologist
SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for Week 36 of 2010, which covers the period from 29 August through 04 September 2010.
1 OPERATION SUMMARY

1.1 Daily average river flows ranged from 80.9 to 108.5 kcfs.  Daily average project tailwater ranged from 7.8’ to 9.9’ msl.  Daily average powerhouse forebay elevation ranged from 75.4’ to 76.2’ msl.  Daily average river temperature ranged from 68 to 70ºF.  Secchi disk measurements ranged from 6.0’ to 7.0’.
1.2 Spill Bays 1 and 18, which provide attraction flow for Cascades Island and B-branch, respectively, are open 6” during daylight hours.  Spill for juvenile fish passage concluded at 2400 on 31 August.  Daily average spill ranged from 1.5 to 64.9 kcfs.
1.3 Unit Operation.  Powerhouse Two is the priority powerhouse. 
1.3.1 Unit 0 went out of service at 0841 on 02 August for five year overhaul.
1.3.2 Unit 9 has remained out of service since 04 August 2008 for turbine rehab.

1.3.3 Unit 11 has remained out of service since 12 June 2008 due to rotor damage.
1.3.4 Unit 13 was out of service from 1300 until 1441 on 31 August for air gap monitor adjustment.
1.3.5 Unit 18 was out of service from 1059 on 30 August until 2356 on 02 September for cooling water pipe troubleshooting.
1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore (WS) fish ladder.  The units may be placed on standby, as necessary, during nighttime hours to float debris.
1.5 Gatewell drawdowns are measured on Wednesdays.
2 ACTIVITIES

2.1 Auxiliary Water System Closures.  The following AWS valves were closed this reporting week:

2.1.1 FV 3-9 was closed for cleaning from 1344 until 1600 on 30 August and from 1420 until 1445 on 02 September.
2.2 STS/VBS inspections.  No STS/VBS inspections occurred this reporting week.
Table 1.  STS and VBS inspections.  
	Unit
	Date
	Run Hours
	Comments

	11
	NA
	OOS
	U11 remains out of service.

	12-18
	N/A
	N/A
	Inspection scheduled for week of 20 September


2.3 Dewatering and Fish Salvage.  The following dewatering/fish salvage activities occurred this week:
2.3.1 Biologist removed five sculpin, one peamouth, and two bass from U18 tail logs on 02 September.  All fish were returned to the river in good condition.  A sturgeon carcass was also recovered.  It was in half, and appeared to have been dead for a few days.
3 RESEARCH  

3.1 CRITFC - Annual chinook, coho, sockeye and steelhead sampling in the AFF continues.  
3.2 NOAA- lamprey.  PIT tagging adult lamprey in the AFF continues.  
3.3 PSMFC- smolt monitoring.  Juvenile fish condition and index monitoring continues.  No fallbacks recorded crossing the separator bars for this reporting week.  Gas Bubble Trauma (GBT) examinations occur two days per week and concluded on 31 August.  Please go to http://www.fpc.org/CurrentDaily/Gbtsum.txt for further details.  Truck releases from McNary Dam, Lower Granite, Little Goose, and Lower Monumental dams continue.
3.4 WDFW- Fish Counting.  Visual counting continues.  Night counts continue.
3.5 WDFW- Tule counting.  Tule counting continues.
3.6 University of Idaho – lamprey.   Radio and PIT tagging adult lamprey in the AFF concluded on 01 September. 
4 FISHWAYS 

4.1 Project biologists inspected 29 August through 04 September.
4.2  Fish Passage Plan Observations.
Table 2.  FPP Out of Criteria items.
	DATE
	LOCATion
	FPP itEMS
	CAUSE
	RESPONSE
	TR#

	BRADFORD ISLAND (BI)

	8/31, 9/2
	B-branch entrance
	Differential <1.0’
	System in manual
	Operators adjusted
	

	CASCADES ISLAND (CI) Nothing to report.

	WASHINGTON SHORE (WS)

	9/2
	18A orifice
	Closed
	Orifice jammed
	Operators adjusted
	


4.2.1 PH1 PCC diffuser FG2-19 is partially open.  Repairs scheduled for 2011/2012 in water work period.  

4.2.2 The SA-24 board that controls the Powerhouse 2 adult ladder components is in need of replacement. A new programmable logic controller (PLC) system has been budgeted for.
4.3 Fishway Temperatures.  Temperatures are taken 01 March through 30 November.

Table 3.  Bonneville Fishway Temperatures

	Date
	WA. Shore average
	BI

average
	AFF average
	SMF average

	8/29
	68.3
	68.6
	68.3
	68.8

	8/30
	67.9
	68.1
	67.9
	68.4

	8/31
	67.2
	67.6
	67.2
	67.8

	9/1
	66.7
	66.9
	66.6
	67.1

	9/2
	66.7
	66.9
	66.6
	67.2

	9/3
	66.8
	67.1
	66.6
	67.2

	9/4
	67.0
	67.1
	66.9
	67.5


4.4 Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week.  
4.5 Adult Fishways. 
4.5.1 Calibration was checked on 03 September.
4.5.2 The Washington Shore fishway entrances remain in manual due to SA-24 board calibration issues.  The operators are adjusting the gates to maintain at least a 13’ opening, as tailwater allows.
4.5.3 Sea lion exclusion bars remain in 11 of 12 floating orifice gates (FOGs).  One FOG remains out for maintenance, with a bulkhead installed in its place.

4.5.4 Sampling at the AFF continues.
4.5.5 Operators concluded the operation of maintaining 0.5’ of head at the PH2 fishway entrances for lamprey passage during night hours on 31 August.  Table Bon-5 of the FPP is being utilized as a guideline.  
4.6 Juvenile Fishways.
4.6.1 DSM2 remains in service.
4.6.2 B2CC was closed for the season at 1549 on 31 August.  The flushing valve remained open for approximately 30 minutes.
4.6.3 The PH1 ice and trash sluiceway (ITS) remains in service.  Chain gate 6C was closed and 7A opened at 0740 on 04 August as outlined in the teletype sent on 29 July (See 7.3).  Chain gate 7A is set to 72’ msl.
4.6.4 Hydrocannon remains in service.
4.6.5 TIES remain installed.  They are scheduled to be removed the week of 07 September.
4.7 Lamprey Fishways.
4.7.1 The Cascades Is. entrance weir and slot filler remain installed.  Cascade Is. LPS remains in service. Scheduled to operate Sunday night through Friday.
4.7.2 The PH2 NDE ramp remains in the lowered position and in service
4.7.3 Bradford Island and WA Shore LPS’s remain in service.
5 CONSTRUCTION
5.1 PH1 Gantry 4.  New crane in FY11.  
5.2 Spillway Gate Repair Pit. Clean-up of hazardous material is in progress.  Shoring of the walls is scheduled to resume when fish counts drop below 1,000.
6 HAZMAT, SPILLS AND CLEANUP   

6.1 On 31 August a small sheen was reported in the tail race of PH2 in the area of Unit 16b and 17a.  Over a 24 hour period a total of 1.5 cups of petroleum products were released.  Upon investigation it was determined that the sheen was emanating from the oil water separator discharge for PH2. The discharge is located in the area of the sheen. 25,000 gallons of oil and water were removed over night; oil recovery was verified as 8% oil / 92% water.  The following day the sheen was no longer present.  The EPA and WA DOE were on-site on 01 September.  NRC Report:  #952642.  WA Emergency Mgmt:  #10-2838.
7 TELETYPES AND OTHER NOTEWORTHY ITEMS
7.1 A teletype sent on 16 July 2009, changes PH2 unit priority to: 12,18,15,17,14,13,16.  This change is effective from 1100 on 16 July 2009 until U11 returns to service.
7.2 A teletype sent on 16 July requests BON project begin the summer spill at 0430 on 21 July until 2400 on 31 August.  Spill 75kcfs during the day and gas cap at night.  Day time hours are defined in table BON-5 of the 2010 FPP.  Spill in accordance with pattern in table BON-16 of the 2010 FPP.  If flows are insufficient, the project is requested to reduce to minimum generation at night and spill the remainder.  The minimum generation requirement is 30 kcfs, and the minimum spill level is 50 kcfs
7.3 A teletype sent on 29 July requests BON project change PH1 unit priority as follows: 1, 3,7,6,2,4,5,8,10,9, effective immediately until further notice.  This moves Unit 7 to third in priority.  This operation is necessary to complete a 100 day operation per PH1 rehab contract.  The project is requested to close automatic chain gate 6C and open manual gate 7A to minimize impacts to fish passage.
7.4 A teletype sent on 18 August requests Bonneville forebay be operated within 1.5’ band, hard constraint, for treaty fishing as follows:  0600 Tuesday, 24 August until 1800 Friday, 27 August;   0600 Tuesday, 30 August until 1800 Friday, 03 September; and 0600 Tuesday, 07 September until 1800 Friday, 10 September.  This is to limit pool fluctuation and reduce wear and tear on equipment.
7.5 A teletype sent on 31 August cancels the teletype sent on 16 July regarding spill (see 7.2).  Spill should stop at 2400 on 31 August, as stated in the 2010 FOP.
7.6 A teletype sent on 01 September requests BON outflows be maintained as low and as constant as possible, requested not to exceed 150 kcfs, from 0700 until 1130 on 06 September.  This operation is requested to reduce river currents for the Cross Channel Swim.

GLEN A. SMITH, P.E.
Operations Project Manager
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